Author's response to reviews

Title: Frequency of sarcopenia and associated factors among hospitalized elderly

Authors:

Bruno P Martinez 1 (brunopmartinez@hotmail.com)
Anne KMS Batista 2 (anne_kmenezes@yahoo.com.br)
Isabela B Gomes 2 (isabelauneb@gmail.com)
Flávia M Olivieri 2 (olivierifau@gmail.com)
Fernanda WR Camelier 3 (fcamelier@uneb.br)
Aquiles A Camelier 1 (aquilescamelier@yahoo.com.br)

Version: 7 Date: 9 March 2015

Author's response to reviews:

Dear reviewers;

We made a small adjustment in Table 2, because had some repeated information.